Author’s response to reviews

Title: Perceptions and intervention preferences of Moroccan adolescents, parents, and teachers regarding risks and protective factors for risky sexual behaviors leading to sexually transmitted infections in adolescents: Qualitative findings

Authors:
Hicham EL KAZDOUH (hicam_kazdouk@yahoo.com)
Abdelghaffar EL-AMMARI (svt_agregation@hotmail.com)
Siham BOUFTINI (siham-bouftini@hotmail.com)
Samira EL FAKIR (elfakirsamira@yahoo.fr)
Youness EL ACHHAB (youness_elachhab@yahoo.fr)

Version: 5 Date: 28 Jul 2019

Author’s response to reviews:

Dear Editor,

Thank you for your reply and comments.

Comments and responses:

Comment:

Editor Comment:

After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

We advise you to seek the assistance of a fluent English-speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract.

Response: thanks for your valuable comments that helped us to improve the quality of this article. Based on your request, we asked for the assistance of a fluent English-speaking colleague
who helped us to revise and correct language. In addition, further attention is paid to the abstract. All the changes made are colored in red.